Author's response to reviews

Title: Osteomyelitis and pyoarthritis caused by localized paracoccidioidomycosis in an immunocompetent patient: a case report

Authors:

Michel S Michelan (michelsmichelan@yahoo.com.br)
Eloy A Fernandes (fernandes_ea@yahoo.com.br)
Leonardo F Freitas (furtadoleofreitas@hotmail.com)
Rodrigo H Ribeiro (centrodeestudosiamspe@yahoo.com.br)
Marilia M Milano (michelsmichelan@yahoo.com.br)
Soraya S Monteiro (sorayasmonteiro@bol.com.br)

Version: 2 Date: 19 March 2012

Author's response to reviews: see over
To Editor in-chief of Journal of Medical Case Reports

From
Dr Leonardo Furtado Freitas, Resident of Radiology Department - Hospital do Servidor Público Estadual (HSPE/IAMSPE), São Paulo (SP), Brazil
Address: Rua Dr. Gabriel dos Santos, 179 apto 41, CEP: 01231-011. São Paulo (SP), Brazil. Telephone: (11) 8931-9955. E-mail: furtadoleofreitas@hotmail.com

Dear Professor

Please find enclosed the manuscript “Osteomyelitis and pyoarthritis caused by localized paracoccidioidomycosis in an immunocompetent patient: a case report” which we are submitting to Journal of Medical Case Reports as an case report article. This case report adds to the medical literature because it is an unexpected/unusual presentation of the disease, in a pediatric immunocompetent patient.

I hereby certify that the manuscript has not been published before nor is it currently under consideration for publication anywhere else and that its publication has been approved by all co-authors, who participated sufficiently to take public responsibility for its content. The author(s) institutions have no conflicts of interest.

Thank you for your attention.

I look forward to hearing from you.

Sincerely yours,

Leonardo Furtado Freitas